Author's response to reviews

Title: Nurse led, primary care based antiretroviral treatment versus hospital care: a controlled prospective study in Swaziland

Authors:

Ciaran P Humphreys (ciaranhumphreys@gmail.com)
John Wright (john.wright@bradfordhospitals.nhs.uk)
John Walley (i.walley@leeds.ac.uk)
Canaan T Mamvura (docmamvura@yahoo.com)
Kerry A Bailey (kerrybaileyjones@googlemail.com)
Sweetness N Ntshalintshali (docmamvura@yahoo.com)
Robert M West (r.m.west@leeds.ac.uk)
Aby Philip (abyshoba@gmail.com)

Version: 7 Date: 28 July 2010

Author's response to reviews:

Dear editor.

I have made formatting changes requested, and additionally several typographical corrections. These are as follows: for consistency use of 'nurse led' throughout rather than on some occasions 'nures-led'; similarly 'antiretroviral' throughout. Consistency of rounding in table 3 (CD4). p4 clarity of sentence re prevalence by addition of 'infected with HIV'. Writing out 53% in full at start of sentence (p 11). For clarity changes to sentence p 14 (‘ill and commence’ to ‘ill who commence’). Minor changes to reference formatting for consistency.

Kind regards,

Ciaran Humphreys